\begin{thebibliography}{10}

\bibitem{T3D-OVERVIEW}
{\sc Adams, D.}
\newblock {Cray T3D System Architecture Overview}.
\newblock Tech. rep., Cray Research Inc., September 1993.
\newblock Revision 1.C.

\bibitem{postal-model}
{\sc Bar-Noy, A., and Kipnis, S.}
\newblock {Designing Broadcating Algorithms in the Postal Model for
  Message-Passing Systems}.
\newblock In {\em Symposium on Parallel Algorithms and Architectures\/} (San
  Diego, June 1992), ACM, pp.~13--22.

\bibitem{adapt-route}
{\sc Berman, P., Gravano, L., Pifarre, G., and Sanz, J.}
\newblock {Adaptive Deadlock- and Livelock-Free Routing with All Minimal Paths
  in Torus Networks}.
\newblock In {\em ACM Symposium on Parallel Algorithms and Architectures\/}
  (San Diego, June 1992), ACM, pp.~3--12.

\bibitem{bokhari91}
{\sc Bokhari, S.}
\newblock {Multiphase complete exchange on a circuit switched hypercube}.
\newblock Tech. Rep. 91-5, ICASE, January 1991.

\bibitem{bokhari92}
{\sc Bokhari, S.~H., and Berryman, H.}
\newblock {Complete Exchange on a Circuit Switched Mesh}.
\newblock In {\em Proc. Scalable High Performance Computing Conference\/}
  (Williamsburg, VA, April 1992), pp.~300--306.

\bibitem{iwarp-general}
{\sc Borkar, S., et~al.}
\newblock {iWarp: An Integrated Solution to High-Speed Parallel Computing}.
\newblock In {\em Proceedings of the Supercomputing Conference\/} (1988),
  pp.~330--339.

\bibitem{iwarp-comm2}
{\sc Borkar, S., et~al.}
\newblock {Supporting Systolic and Memory Communication in iWarp}.
\newblock Tech. Rep. CMU-CS-90-197, Carnegie Mellon University, School of
  Computer Science, 1990.
\newblock Revision of a paper that appeared in the 17th Annual Intl. Symposium
  on Computer Architecture, Seattle, 1990, pp. 70-81.

\bibitem{logp-extended}
{\sc Culler, D., Karp, R., Patterson, D., Sahay, A., Schauser, K., Santos, E.,
  Subramonian, R., and {von Eicken}, T.}
\newblock {LogP: towards a realistic model of parallel computation}.
\newblock Tech. Rep. UCBC 92-713, Univ. of California, Berkeley, 1992.
\newblock expanded version of paper in 4th Symp. on PPoPP.

\bibitem{connset}
{\sc Feldmann, A., Stricker, T., and Warfel, T.}
\newblock {Supporting sets of arbitrary connections on iWarp through
  communication context switches}.
\newblock In {\em ACM Symposium on Parallel Algorithms and Architectures\/}
  (Schloss Velen, Westfalia, Germany, July 1993).

\bibitem{hpf}
{\sc High Performance Fortran Forum}.
\newblock {\em {High Performance Fortran Language Specification Version 1.0.}},
  May 1993.

\bibitem{hinrichs-thesis-work}
{\sc Hinrichs, S.}
\newblock {Compiler Resource Management for Con\-nec\-tion\--Based
  Communication}.
\newblock Internal document, 1994.

\bibitem{aapc-tr}
{\sc Hinrichs, S., Kosak, C., O'Hallaron, D., Stricker, T., and Take, R.}
\newblock {An Architecture for Optimal All-to-All Personalized Communication}.
\newblock Tech. Rep. CMU-CS-94-140, Carnegie Mellon University, School of
  Computer Science, 1994.

\bibitem{horie92}
{\sc Horie, T., and Hayashi, K.}
\newblock {All-to-All Personalized Communication on a Wrap-around Mesh}.
\newblock In {\em Proceedings of CAP Workshop\/} (Canberra, Austrailia,
  November 1991).

\bibitem{AP1000-NETWORK}
{\sc Horie, T., Ishihata, H., and Ikesaka, M.}
\newblock {Design and implementation of an interconnection network for the
  AP1000}.
\newblock In {\em Proc. IFIP World Computer Congress\/} (1992), vol.~I,
  Information Processing, pp.~555--561.

\bibitem{PARAGON-XPS}
{\sc Intel Corp.}
\newblock {\em Paragon X/PS Product Overview}, March 1991.

\bibitem{johnsson89}
{\sc Johnsson, S.~L., and Ho, C.-T.}
\newblock {Optimum Broadcasting and Personalized Communication in Hypercubes}.
\newblock {\em IEEE Transactions on Computers 38}, 9 (September 1989),
  1249--1268.

\bibitem{CM5-NETWORK}
{\sc Leiserson, C., Abuhamdeh, A., Douglas, D., Feynman, C., Ganmukhi, M.,
  Hill, J., Hillis, D., Kuszmaul, B., St.Pierre, M., Wells, D., Wong, M., Yang,
  S., and Zak, R.}
\newblock {The Network Architecture of the Connection Machine CM-5}.
\newblock In {\em Symposium on Parallel Algorithms and Architectures\/} (San
  Diego, June 1992), ACM, pp.~272--285.

\bibitem{MPI-DRAFT}
{\sc The Message Passing Interface Forum}.
\newblock {\em Draft Document for a Standard Message Passing Interface},
  November 1993.

\bibitem{scott91}
{\sc Scott, D.~S.}
\newblock {Efficient All-to-All Communication Patterns in Hypercube and Mesh
  Topologies}.
\newblock In {\em The Sixth Distributed Memory Computing Conference
  Proceedings\/} (1991), pp.~398--403.

\bibitem{IBM-SP1}
{\sc Snir, M.}
\newblock {Scalable Parallel Computing - The IBM 9076 Scalable
  POWERParallel-1}.
\newblock In {\em ACM Symposium on Parallel Algorithms and Architectures\/}
  (June 1993), ACM, p.~42.

\bibitem{FX-COMMUNICATION}
{\sc Stichnoth, J., O'Hallaron, D., and Gross, T.}
\newblock {Generating Communication for Array Statements: Design,
  Implementation, and Evaluation}.
\newblock {\em Journal of Parallel and Distributed Computing\/} (1994).
\newblock to appear.

\bibitem{stricker-routing}
{\sc Stricker, T.}
\newblock {Message Routing on Irregular 2D-Meshes and Tori}.
\newblock In {\em Proceedings of the 6th Distributed Memory Computing
  Conference\/} (Portland, OR, Apr. 1991), pp.~170--177.
\newblock Also appeared as Technical Report CMU-CS-91-109, Carnegie Mellon
  School of Computer Science.

\bibitem{msg-t3d-tr}
{\sc Stricker, T., Stichnoth, J., O'Hallaron, D., Hinrichs, S., and Gross, T.}
\newblock {Decoupling Communication Services for Compiled Parallel Programs}.
\newblock Tech. Rep. CMU-CS-94-139, Carnegie Mellon University, School of
  Computer Science, 1994.

\bibitem{fx-ppopp}
{\sc Subhlok, J., Stichnoth, J., O'Hallaron, D., and Gross, T.}
\newblock {Exploiting Task and Data Parallelism on a Multicomputer}.
\newblock In {\em Fourth ACM SIGPLAN Symposium on Principles and Practice of
  Parallel Programming\/} (San Diego, CA, May 1993).

\bibitem{pvm}
{\sc Sunderam, V.~S.}
\newblock {PVM: a Framework for Parallel Distributed Computing}.
\newblock {\em Concurrency: Practice and Experience 2}, 4 (December 1990),
  315--39.

\bibitem{take87}
{\sc Take, R.}
\newblock {A Routing Method for All-to-all Burst on Hypercube Networks}.
\newblock In {\em Proc. 35th National Conference of Information Processing
  Society of Japan\/} (1987), pp.~151--152.
\newblock In Japanese.

\bibitem{take91}
{\sc Take, R., Noguchi, Y., and Yokota, H.}
\newblock {An Architecture for Parallel Database Computing}.
\newblock In {\em Transputing '91\/} (1991), pp.~1--14.

\bibitem{var92}
{\sc Varvarigos, E.~A., and Bertsekas, D.~P.}
\newblock {Communication algorithms for isotropic tasks in hypercubes and
  wraparound meshes}.
\newblock {\em Parallel Computing 18\/} (1992), 1233--1257.

\end{thebibliography}

\section{Using a standard message passing system for AAPC}
\label{msg-sect}

Most distributed memory machines offer message passing communication.
Message passing ensures door-to-door delivery, so messages are
eventually delivered to the destination without further program
action.  Common models of message passing quantitatively characterize
the message passing system by key parameters like message throughput
and processing overhead and suggest that the programmer does not rely
on the particular details of the target
architecture\cite{logp-extended,postal-model}.  Figure~\ref{mp-code}
shows a simple message passing AAPC program.

\begin{figure}[bht]
\hspace*{\fill}\fbox{\begin{minipage}[t]{\caplen}
{\helv\begin{tabbing}
---\=---\=---\=---\=---\=\kill
 1\>{\bh for} i {\bh from} 0 {\bh to} NumberOfProcessors-1 {\bh do}\\
 2\>\>NBSendMessage(Destination(i), Route(i), DataBlock[i])\\
 3\>{\bh for} j {\bh from} 0 {\bh to} NumberOfProcessors-1 {\bh do}\\
 4\>\>NBReceiveMessage(DataBlock[j])
\end{tabbing}}
\end{minipage}}\hspace*{\fill}

\hspace*{\fill}\begin{minipage}[t]{\caplen}
\caption{Message passing AAPC code. We assume NBSendMessage and
NBReciveMessages are buffered, non-blocking primitives offered by the
message passing library.}
\label{mp-code}
\end{minipage}\hspace*{\fill}
\end{figure}

Both phased AAPC and message passing AAPC decompose the AAPC pattern
into messages and route them through the network. However, in message
passing AAPC, a node passes a batch of $n$ messages to the network.
The wormhole router then routes these messages to $n$ different
processors.  Since the network is an independent subsystem with no
information about the system-wide communication structure, the router
uses a simple, greedy scheduling strategy.  Whenever a requested
communication link becomes available, a message will proceed.

Routes between nodes in the AAPC phases described in
Section~\ref{patterns-sect} are the same routes generated by the
simple e-cube torus router. By following the communication schedule
for the phased AAPC, a basic message passing library should recreate
the same AAPC phases and achieve similar performance.  However, not
all nodes will finish each phase at the same time even if the messages
are the same size, because not all nodes are scheduled to send
messages in each phase. On iWarp, the performance of unsynchronized
nodes following the phased AAPC schedule and using the basic message
passing package was about the same as the performance following a
random schedule.

Since message passing libraries do not rely on the communication
structure for scheduling or routing, message passing is more adaptable
in situations where the communication structure is sparse, changing,
or unknown.

%\nocite{MPI-DRAFT}

\subsection{The iWarp message passing system}
\label{msgpass-sect}
On iWarp, the basic message passing routers are based on a reverse
e-cube scheme\cite{stricker-routing}.  All routes start in the
X-direction, eventually turn corners, and continue in the
Y-direction. The queues can be assigned to multiple pools and the
routers can use date-lines to break circular dependencies, avoiding
routing deadlocks.  Many other commercial, distributed memory systems
use similar simple e-cube or reverse e-cube routing schemes. While
most routers are fixed in the communication hardware, the iWarp system
enables a variety of virtual channel configurations and permits
different router software to determine the precise route of every
message before it is sent.

Previous work on mesh routers has shown that the non-adaptive e-cube
routers can have severe performance limitations, due to imbalanced
congestion.  Several advanced, adaptive wormhole routers have been
proposed for machines with virtual channels\cite{adapt-route}, but
only few of them have been implemented in hardware. We have tested
some of these advanced routers with our iWarp message passing system.
Unfortunately the advanced methods delivered only up to 30\% of
improvement over the basic e-cube scheme, and with only two resource
pools, these advanced methods could not use the torus routing
capability.  Faced with the limitation of only two pools, we chose the
torus routing over the randomized, more adaptive router, so our
measurements in Section \ref{measure-sect} are based on a
deterministic e-cube router with the torus routing capability.

To compare a specialized AAPC architecture with a general message
passing system, we are most concerned with the issues of routing and data
transfer. For this reason, we did not use a standard message passing
interface such as PVM\cite{pvm} or MPI\cite{MPI-DRAFT}. These standard interfaces have software
overheads resulting from error and protection checking in the
operating system, buffer management, and standardized synchronization
semantics, which result in copying data. These aspects of message
passing libraries are an important issue of research, but for this
study they would have obscured the architectural insights into
the nature of an all-to-all communication.  Instead of a standard
library, we use the deposit message passing library written for the Fx
compiler\cite{fx-ppopp}, based on the deposit model\cite{msg-t3d-tr}.  The
deposit library allows direct deposit of incoming data to its final
destination.  At the time it is sent, a message is guaranteed to
encounter a receiver that is ready to extract it from the network
immediately, minimizing buffering overheads and message congestion at
the receiver.
%  The deposit model is similar to the {\em sender
%buffered} message passing introduced in~\cite{connset}.

The measured overhead per transferred message is constant at around
$400$ cycles ($20 \mu s$). The network latency is proportional to the
network diameter $n$ with a cost of 2 to 4 cycles per hop.
Large data blocks are injected as single messages into the network,
and once data is flowing, the transfer rate is limited only by the
link bandwidth of $40$ MBytes/sec.  The same maximal transfer rate
applies to all of our AAPC implementations.








